Author's response to reviews

Title: Protocol for population testing of an Internet-based Personalised Decision Support system for colorectal cancer screening

Authors:

Carlene J Wilson (carlene.wilson@flinders.edu.au)
Ingrid H K Flight (ingrid.flight@csiro.au)
Ian T Zajac (ian.zajac@csiro.au)
Deborah Turnbull (deborah.turnbull@adelaide.edu.au)
Graeme P Young (graeme.young@flinders.edu.au)
Stephen R Cole (steve.cole@health.sa.gov.au)
Tess Gregory (tess.gregory@adelaide.edu.au)

Version: 5 Date: 14 September 2010

Author's response to reviews: see over
14th September 2010

BMC Series Editorial Production Team

Dear Sir/Madam

**MS 1558167435405345:** Wilson, Flight, Zajac et al: *Protocol for population testing of an internet-based Personalised Decision Support system for colorectal cancer screening*

Thank you for your email of 10 September 2010, addressed to Ingrid Flight, in which you requested that we proof read the final, pre-publication version of our article.

We have made some small typographical amendments to the manuscript, and additionally have taken the opportunity to change some of the text to reflect the current situation. Details of these latter changes are:

1. **Page 11, second paragraph**
This paragraph has been amended to reflect a change in protocol whereby participants will now be contacted by letter 1–2 weeks after receipt of their completed FOBT, or 12 weeks following the invitation to screen for those who haven’t returned their kit during this period, and requested to complete an Endpoint Survey.

2. **Page 13, second paragraph (Time plan for study)**
This paragraph has been amended to reflect the current situation, ie that at September 1st, N=3240 (100% of target) had been recruited into the study.

3. **Table 2, phase 3.1**
The introductory description of this phase has been amended to reflect the change in protocol described in point 1 above.

We trust that the revised manuscript meets your requirements.

Thank you for your attention.

[Signature]

Professor Carlene Wilson BA (Hons) PhD MBA